Author’s response to reviews

Title: Plasma HER2ECD a promising test for patient prognosis and prediction of response in HER2 positive breast cancer: Results of a randomized study - SAKK 22/99

Authors:

Serenella Eppenberger-Castori (sere.eppenberger@unibas.ch)
Dirk Klingbiel (dirk.klingbiel@roche.com)
Thomas Ruhstaller (thomas.ruhtaller@kssg.ch)
Daniel Dietrich (daniel.dietrich@sakk.ch)
Daniel Rufle (danielalexander.rufle@aprentas.com)
Karin Rothgiesser (karin.Rothgiesser@sakk.ch)
Olivia Pagani (opagani@bluewin.ch)
Beat Thürlimann (beat.thuerlimann@kssg.ch)

Version: 4 Date: 29 Jan 2020

Author’s response to reviews:

Dear Mr. Jones, dear Mrs. Rice and all Editors in charge

Thank you so much for the opportunity of revising the statement regarding the Availability of data and materials. We have now added that they can be available upon reasonable request.

In the hope to have eventually fulfilled all requests,

sincerely

Serenella Eppenberger